

Endpoints selection in registration clinical trials and the needs of real-world clinical practice with the example of anti-VEGF therapy in neovascular agerelated macular degeneration

© Sergey Yu. Astakhov¹, Svyatoslav L. Plavinskii²

¹ I.P. Pavlov First Saint Petersburg State Medical University, Saint Petersburg, Russia;

² I.I. Mechnikov North-Western State Medical University, Saint Petersburg, Russia

The issues of endpoints selection for regulatory requirements and real-world clinical practice using the example of anti-VEGF therapy in neovascular age-related macular degeneration (nAMD) are discussed in the article. New technologies (optical coherent tomography) introduction are shown to change clinical practice but not regulatory requirements on the endpoints. In the same time for regulatory purpose clinical trials design is changed from superiority to non-inferiority. The changes in the approach to primary endpoint selection are not anticipated due to regulator's conservatism but there is a requirement to the comparison with best treatment alternative (i.e. same class comparator in case of anti-VEGF therapy) due to ethical reasons. To satisfy real-world clinicians need, the secondary endpoints are analyzed, but multiple testing problem appears. Statistical methods developed in recent years allow using specified comparison to be made without inflating Type I error. HAWK and HARRIER clinical trials demonstrated an example how superiority of brolucizumab over aflibercet on anatomical endpoints was reliably found.

Keywords: clinical trials endpoints; clinical trial ethics; multiple comparisons; hierarchical statistical testing; evaluation of anti-VEGF therapy in neovascular AMD; anatomical endpoints.

To cite this article:

Astakhov SYu, Plavinskii SL. Endpoints selection in registration clinical trials and the needs of real-world clinical practice with the example of anti-VEGF therapy in neovascular age-related macular degeneration. *Ophthalmology Journal*. 2021;14(1):25-33. DOI: https://doi.org/10.17816/0V63571

Received: 11.01.2021



Accepted: 10.02.2021

Published: 23.03.2021

Особенности выбора конечных точек в регистрационных исследованиях и потребности реальной клинической практики на примере анти-VEGF-терапии неоваскулярной возрастной макулярной дегенерации сетчатки

© С.Ю. Астахов¹, С.Л. Плавинский²

¹ Федеральное государственное бюджетное образовательное учреждение высшего образования

«Первый Санкт-Петербургский государственный медицинский университет имени академика И.П. Павлова»

Министерства здравоохранения Российской Федерации, Санкт-Петербург, Россия;

² Федеральное государственное бюджетное образовательное учреждение высшего образования

«Северо-Западный государственный медицинский университет имени И.И. Мечникова» Министерства здравоохранения Российской Федерации, Санкт-Петербург, Россия

На примере анти-VEGF-терапии в лечении неоваскулярной возрастной макулярной дегенерации сетчатки продемонстрированы проблемы выбора конечных точек для требований регуляторных органов и реальной клинической практики. Показано, как появление новых технологий (оптической когерентной томографии) стимулирует изменение в клинической практике, однако не в конечных точках, измеряемых в исследованиях, с изменением дизайна с превосходства на не меньшую эффективность. Регуляторные органы не могут допустить смены конечной точки ввиду консерватизма, но по этическим причинам требует сравнения с наилучшим альтернативным препаратом, в случае анти-VEGF-терапии — из того же класса. Для удовлетворения потребностей клиницистов организаторы исследований проверяют также вторичные точки, но возникает проблема множественного тестирования. Разработанные в последние годы статистические методы позволяют заранее специфицировать сравнения и убрать опасность увеличения ошибки I типа. Приведен пример исследований HAWK и HARRIER, как было установлено превосходство бролуцизумаба над афлиберцептом по анатомическим конечным точкам.

Ключевые слова: конечные точки в клинических испытаниях; этика клинических испытаний; множественные сравнения; иерархическое статистическое тестирование; оценка анти-VEGF-терапии неоваскулярной возрастной макулярной дегенерации сетчатки; нВМД; анатомические конечные точки.

Как цитировать:

Астахов С.Ю., Плавинский С.Л. Особенности выбора конечных точек в регистрационных исследованиях и потребности реальной клинической практики на примере анти-VEGF-терапии неоваскулярной возрастной макулярной дегенерации сетчатки // Офтальмологические ведомости. 2021. Т. 14. № 1. С. 25–33. DOI: https://doi.org/10.17816/0V63571

Рукопись получена: 11.01.2021

Рукопись одобрена: 10.02.2021

Опубликована: 23.03.2021



The planning and the organization of clinical trials are labor-intensive and complicated matter, whereby being highly expensive [1]. Researchers have to take into consideration a whole bunch of aspects applied to organizers, from ethical principles of study conduct to requirements of licensing/regulating authorities to evidence level concerning efficacy and safety. One of the problems could be the choice of the clinical trial's endpoint on which its sponsor will rely determining the number of volunteers to be involved in the trial, its duration, and special aspects of statistical data processing. The understanding of how the problems facing the researchers of anti-VEGF medications which are used to treat neovascular age-related macular degeneration (nAMD) are resolved, is necessary to adequately estimate the body of evidence created for new medications.

The aim of present article is to present the features of new anti-VEGF molecules development and to show why the anatomical endpoints (such as fluid in the retina (intraretinal, subretinal) and that under the retinal pigment epithelium; central retinal thickness (CRT); disease activity determined by the loss in best corrected visual acuity (BCVA) of more than 5 letters and by new macular hemorrhages [2–4]) were used as secondary endpoints.

PRESENTATION OF BASIC MATERIAL

When planning a clinical trial, the endpoints, namely variables determining efficacy and safety, are divided into primary (basic) and secondary ones. Usually, there are few primary points. They are used to calculate the required sample size for the trial and to conclude on efficacy and safety. Secondary points are used as supporting ones, and if the study protocol requires superiority or non-inferiority in efficacy (see below), regulating authorities may not give their consent to accept data where the efficacy is determined from secondary points and require to carry out one more clinical trial.

At the same time, at the development of anti-VEGF medications, researchers had to face some difficulties influencing this program. The point is that the development of anti-VEGF medications went in parallel with the development of technological methods for determination of anatomical parameters, first of all of optical coherence tomography (OCT). First registered anti-VEGF medication for nAMD treatment – pegaptanib sodium – was approved by the American Food and Drug Administration (FDA) in 2004, 11 years after the appearance of first publications about the OCT use to obtain retinal images. At that time, according to the analysis of the American National Medical Library database, there were only 17 articles published on the problem of the OCT use in nAMD (Table 1).

Obviously, the researchers of first medications had only functional evaluation methods available, namely BCVA estimation. In 2005, the results of a study were published dealing with anti-VEGF monoclonal antibody (bevacizumab) use in nAMD, which was used as offlabel injections [5]. The main endpoint was certainly the visual acuity. Several clinical trials on the intravitreal bevacizumab administration were carried out [6, 7]. They were retrospective by design and did not have the degree of control on samples collection quality and procedure performance which characterizes prospective clinical studies.

In 2006, FDA licensed ranibizumab for nAMD treatment. It was done on the basis of studying its comparison to sham intravitreal injections (MARINA [8] - placebocontrolled study) and to photodynamic therapy (ANCOR [9]). Data of a phase I/II clinical trial (FOCUS [10]) were also used, where on the photodynamic therapy background, either ranibizumab or sham injections were performed. Visual acuity indicators were used as endpoints, and ranibizumab superiority was shown. It has to be stressed that the comparison was carried out with sham injections and with photodynamic therapy and not with anti-VEGF therapy (pegaptanib), because authorizations were issued with two-year interval and correspondingly, taking into consideration longer duration of registration clinical studies at the moment of ranibizumab trials, pegaptanib was not available for extensive use in the framework of multicenter registration studies. Studies carried out later to indirectly compare ranibizumab with pegaptanib allowed assuming higher ranibizumab efficacy, in spite authors pointed out to different populations included into ranibizumab trials in comparison to pegaptanib trials [11]. Nevertheless, large agencies evaluating technologies in healthcare (for example, British NICE) did not recommend using pegaptanib for nAMD treatment¹.

Taking into account the load upon the healthcare system, which was related to both with the price of

Таблица 1. Количество публикаций в системе PubMed об использовании OKT при нВМД (поиск осуществлен на https:// pubmed.ncbi.nlm.nih.gov по сочетанию AMD [TW] AND OCT [TW] 05/01/2021)

Years	Number of publications	During a year, on average	
1993–2004	17	1.5	
2005–2011	269	38.4	
2012–2020	987	109.7	

¹ NICE. Ranibizumab and pegaptanib for the treatment of age-related macular degeneration. Technology appraisal guidance. Published: 27 August 2008. www.nice.org.uk/guidance/ta155.

Table 1. Number of publications in PubMed database on OCT usein wet AMD (search was performed on 05/01/2021 at https://pubmed.ncbi.nlm.nih.gov with keywords AMD [TW] AND OCT[TW])

innovation medications and with the monthly regimen of therapy, ophthalmologists began to concern themselves with increasing interval between injections without any harm to the patient. Anatomical points could be of assistance, which allowed in advance objectively evaluating worsening of the condition or slowing down of the disease progression without any substantial visual acuity loss. This increased the interest to the OCT use and to the determination of the relationship between anatomical changes and disease progression. Between 2004 and 2011, already 269 articles were published on the OCT use in nAMD. The technology itself continued to develop, becoming more accessible for clinicians.

To 2011, the clinical development program of aflibercept for nAMD treatment was accomplished. The studies continued from middle 2000s, and the endpoint was visual acuity again. A licensed effective medication ranibizumab already existed, and for ethical reasons, the comparison had to be made with it. As it is easier to prove superiority over placebo than over another effective treatment, VIEW 1 and 2 trials were based on the non-inferiority design [12]. At the non-inferiority design, there are no attempts to show that the new medication is more effective than the existing therapy or placebo. The objective is to demonstrate that it is not worse. The frontier of lower efficacy is established, which cannot be overlapped by the 95% confidence interval's lower limit [13]. Thus, a statistical warranty is given that the preparation would not approach placebo and be effective. On a formal level, it may be a little bit better or a little bit worse. "Non-inferiority" designs became popular over recent years due to the appearance of new preparations and of questions for the ethics of placebo control organization in presence of effective treatment. As a result, if in 2005 in PubMed, there were less than 100 publications of non-inferiority trials, in 2005, there were already almost 600.

At the same time, there was a growing understanding of the importance of OCT and of anatomical points measurement for evaluation of the treatment's efficacy. The number of publications concerning the OCT use in nAMD increased practically according to the exponential curve between 2011 and 2020, 987 articles were published (about 110 in one year, or just below 10 in one month). During this period of time, the understanding of the importance of anatomical points to monitor the therapy appeared. At the same time, questionnaire data of the "2019 EURETINA Clinical Trends Survey Outcomes"² show that very few ophthalmologists use regular monthly injections (5% in the Western Europe and 11% - in the Eastern Europe), the rest uses PRN (pro re nata, that is "as needed", or under present circumstances, – 30% ophthalmologists of the Eastern Europe) or T&E (Treat and extend - 57% ophthalmologists of the Eastern Europe in the survey) regimens, in which anatomical indicators (namely, the presence of fluid) play an important role.

It became obvious for example that preservation and appearance of intraretinal/subretinal fluid (IRF/SRF)is associated with more significant visual acuity loss: loss of 15 letters in 2.8% of patients with new IRF and the same number of patients with new SRF. In total, a loss of more than 5 letters is noted in 29.6% of patients with recurrent SRF and in 33.9% – with recurrent IRF. In patients without fluid, this amount was equal to 16.6% only (ranibizumab therapy was conducted according to T&E protocol) [14].

Later on, a clinical development of one more anti-VEGF monoclonal antibody – brolucizumab took place. From the very start of the program, one could understand that in all studies the main endpoint is the visual acuity, because it was used in previous trials and is ranged by regulatory authorities as a "real" endpoint, i. e. an indicator, which measures something important for the patient. Brolucizumab studies begin with phase I/II trials, in which a search for acting dose and preliminary efficacy evaluation are performed. In the SEE study [15], there were two parts: on the first stage 0.5, 3.0 or 4.5 mg brolucizumab or 0.5 mg ranibizumab were injected, on the second – 0.5, 3.0 or 6.0 mg brolucizumab or 0.5 mg ranibizumab.

Non-inferiority in efficacy between brolucizumab 4.5 and 6.0 mg as compared to ranibizumab. CRT changes were 23 μ m more (90% CI = -9.28...54.99) in the group, which received brolucizumab at a dose of 4.5 mg, and 19 μ m more (90% CI = -9.0...47.8) in the group, which received 6 mg of brolucizumab, than in the ranibizumab group.

The OWL study was already only II phase trial, in which brolucizumab was compared to ranibizumab when using intravitreal injections or microinfusions [15]. There were 52 patients in total, randomized at a ratio (thus in the ranibizumab group few patients happened to be included – 12 only). Brolucizumab treatment was performed as: 1.2 mg/10 mcl injection; 0.6 mg/10 mcl injection; .0 mg /8.3 mcl microinfusion; 0.5 mg/8.3 mcl microinfusion. Ranibizumab: intravitreal injection 0.5 mg/50 mcl. In both groups a positive treatment result was noted.

One more II phase study was accomplished – OSPREY [16], it was a 56-week study comparing the efficacy of 6 mg brolucizumab and 2 mg aflibercept intravitreal injections. There were 89 participants in total receiving a therapy following the pattern: 3 injection intervals every 4 weeks, then every 8 weeks, and later on every 12 weeks in the brolucizumab group and 8 weeks in the aflibercept group. The so-called posthoc-analysis included the evaluation of CRT changes and of SRF/IRF presence. CRT changes on Week 12 were -197 and -189 μ m, on Week 40 – -198 and -178 μ m, correspondingly. The number of eyes with SRF/IRF was 9.3 μ 20.9% on Week 12 and 14.6 and 32.5% on Week 40 in brolucizumab and aflibercept

² The Interim 2019 EURETINA Clinical Trends Survey Outcomes // The great fluid debate – Retina Today. https://retinatoday.com/pdfs/1219_ insert4.pdf. Access regimen: 27.11.2020.

groups. Obviously, one could predict, that brolucisumab had an edge over aflibercept concerning anatomical points. But as mentioned before, regulatory authorities would not accept a transition to a new endpoint.

There is also one more problem, encountered when analyzing several groups in studies, - this is the problem of multiple comparisons [17]. At its simplest, its description looks like following: every time, when we perform group comparisons, we decline the zero hypotheses (about the nondiversity) at 5% level (0.05). Correspondingly, in 5% of studies, comparing two equally effective interventions, there is a risk to acknowledge superiority in one of the groups, although this is not true. If an analysis of 3 points is performed, the probability of false conclusion would be already not 5%, but 14.3%. This is why the multiplicity of comparison has to be taken into account somehow. The most simple could be the Bonferroni method, which simply lowers the level of significance to a number of comparisons – if they are 5. The limit level would be equal to 1%. Therefore, to assure the possibility of significant efficacy comparison based on changes of anatomical points, the protocol was very carefully adjusted from the statistical point of view.

And now some words about registration studies HAWK and HARRIER [18]. By design, it were two-year multicenter randomized double-blind trials of "non-inferior efficacy" with active control. As a control анти-VEGF medication, aflibercept was chosen. The HAWK trial: 1082 nAMD patients were randomized as 1:1:1 into three groups, receiving brolucizumab at a dose of 3 mg, brolucizumab – 6 mg or aflibercept – 2 mg as intravitreal injections. The HARRIER trial: 743 patients were randomized as 1:1 into two groups, receiving intravitreal injections of brolucizumab at a dose of 6 mg or aflibercept at a dose of 2 mg. HAWK was carried out in the Northern America, Latin America, Japan, Australia, New Zealand, and Israel. HAR-RIER – in EC countries, the Middle East, Asia, and Russia.

The primary outcome was the change of best corrected visual acuity (BCVA) from the study start up to Week 48. It has to be noted that the duration of the study was 96 weeks, and during the whole period, the follow-up and treatment of patients continued, but there couldn't be any new primary endpoint at the late stages of the trial. As patients of different age groups and with different visual acuity, the analysis included correction with a statistical model and calculation of BCVA difference between Week 48 and baseline data. Notably, the statistical model adjusted differences related to different baseline visual acuity and age (so-called predicted marginal means, or leastsquares means [LSMEANS]). As non-inferiority margin, 4 signs were chosen³. The results appeared as follows:

³ Calculations of the lower efficacy limit were performed according to data on aflibercept efficacy in comparison with ranbizumab, and of ranibizumab in comparison with placebo provided no less than 50% of the effect in comparison with placebo was preserved. HAWK -0.16 (95% CI [-2.13; 1.80]) (brolucizumab at a dose of 6 mg in comparison with aflibercept), HARRIER -0.70 (95% CI [-2.39; 1.00]). Lowest values of 95% confidence interval were higher than the non-inferiority margin, and this means that the primary goal of the study - obtaining non-inferiority evidences - was reached. It is important because in such complex designs, a hierarchical testing is used, and if the results on primary outcome were not positive, the following testing would be inconclusive. Such tough approach to the evaluation of clinical study results allows creating models, which allow to answer the question: what if in clinical study brolucizumab would be compared not with active control, but with placebo? With this in mind, H. Agostino, et al. [19] carried out a comparison of HAWK/HARRIER data with data of model effect on placebo. The model was created based on ANCOR, MARINA, PIER, EXCITE data и validated on HARBOR study data [20]. It was taken into account that age and baseline visual acuity play a prognostic role. According to the model, the effect from brolucizumab use versus placebo was equal to +22 signs by Week 48, and +28 signs by Week 96.

The main objective of the present study was to show, want efforts were under way to ensure the level of evidence of comparing anatomical points by brolucizumab and aflibercept therapy. Based on results of prervious studies on brolucizumab efficacy, following indicators were chosen as main indices of testing: CRT change, absence of IRF/SRF, and disease activity. These three indicators had to be taken into consideration to preserve on the predetermined level the type 1 errors, or as they say to prevent "false discoveries". This was made only in HAWK trial (statistical analysis in HAWK was created taking into account HARRIER [18]). The authors of the statistical analysis plan in the study decided to choose the method of type 1 error (α -error) between indices level separation (global test for superiority). Because they intended to use single tail tests, they had to divide the error level of 2.5% (0.025, or conventional twotailed significance level of 0.05, divided into 2). The authors decided to divide the error level as follows: CRT change -0.005; absence of IRF/SRF - 0.01; disease activity - 0.01. It has to be noted that the protocol authors considered the disease activity and absence of IRF/SRF more important than CRT changes, because to prove the superiority in this point was more difficult. Further on, multiple hierarchal tests were done, as for example for CRT changes:

1) brolucizumab at a dose of 6 mg vs aflibercept at Week 16;

2) brolucizumab at a dose of 6 mg vs aflibercept at Week 48;

3) brolucizumab at a dose of 6 mg vs aflibercept for median of Weeks 36, 40, 44, 48;

4) brolucizumab at a dose of 3 mg vs aflibercept at Week 16;

5) brolucizumab at a dose of 3 mg vs aflibercept at Week 48;

29

6) brolucizumab at a dose of 3 mg vs aflibercept for median of Weeks 36, 40, 44, 48.

Here 6 hypotheses are tested sequentially, from 1^{st} to 6^{th} , at that if even one hypothesis in the sequence happens to be accepted (the *p* value happens to be higher than the aforesaid limit), further hypotheses are not tested (Table 2), this approach has a name of gatekeeping method [21].

Then, reliable information was obtained on the superiority of brolucizumab therapy at a dose of 6 mg by anatomical endpoints, first of all, by IRF/SRF presence and CRT, because the evaluation for the disease activity estimation was performed only on Week 16 due to the change of disease activity determination at more advanced terms. Attention must be paid to extremely low p values in HAWK study, which, taking in mind the preliminary formulation in the protocol protects against false discoveries. It is interesting that in the publication on the results of 96-weeks trials HAWK μ HARRIER [24], the authors found it necessary to point out that p estimations are descriptive (because the analysis on Week 96 was not foreseen), however data on anatomical points were also highly significant and spoke well for brolucizumab.

CONCLUSION

The development of science and technological basis of medicine leads to the appearance of new markers to monitor and modify the therapy. However easily understandable conservatism of the regulatory system in combination with

Table 2. Evaluation of superiority hypothesis in HAWK and HARRIER trials (supplementary materials to article by P.U. Dugel, et al [18])
Таблица 2. Оценка гипотез превосходства в исследованиях HAWK и HARRIER (дополнительные материалы к статье P.U. Dugel и соавт. [18])

Indicator	Endpoint evalua- tion at a period	Brolucizumab dose, mg	Local significance level	<i>p</i> for superiority (single tail test)	Significance according to testing procedure
CRT (HAWK)	Week 16	6	.005	<.001	Yes
	Week 48	6	.005	.001	Yes
	Weeks 36-48	6	.005	.008	No, <i>p</i> > .005
	Week 16	3	.005	.016	Previous hypothesis was not rejected
	Week 48	3	.005	.008	Previous hypothesis was not rejected
	Weeks 36-48	3	.005	.018	Previous hypothesis was not rejected
CRT (HARRIER)	Week 16	6	_	<.001	NA
	Week 48	6	_	<.001	NA
	Weeks 36-48	6	_	<.001	NA
Presence of IRF/ SRF (HAWK)	Week 16	6	.01	<.001	Yes
	Week 48	6	.01	<.001	Yes
	Weeks 36-48	6	.01	.001	Yes
	Week 16	3	.01	.003	Yes
	Week 48	3	.01	.002	Yes
	Weeks 36-48	3	.01	.057	No, <i>p</i> > .01
Presence of IRF/ SRF (HARRIER)	Week 16	6	_	<.001	NA
	Week 48	6	_	<.001	NA
	Weeks 36-48	6	_	<.001	NA
Presence of active disease (HAWK)	Week 16	6	.01	.001	Yes
	Week 16	3	.01	.033	No, <i>p</i> > .01
Presence of active disease (HARRIER)	Week 16	6	_	.002	NA

Note. CRT - central retinal thickness, IRF/SRF - intraretinal/subretinal fluid.

ethical demands to interventions in the control group creates a situation when such markers cannot be primary indicators of efficacy. In such conditions, a careful planning of the statistical analysis allows to avoid the danger of "false discoveries" and to affirm that the revealed superiority is

REFERENCES

1. DiMasi JA, Grabowski HG, Hansen RW. Innovation in the pharmaceutical industry: New estimates of R&D costs. *J Health Econ.* 2016;47:20–33. DOI: 10.1016/j.jhealeco.2016.01.012

2. Arnold JJ, Markey CM, Kurstjens NP, Guymer RH. The role of sub-retinal fluid in determining treatment outcomes in patients with neovascular age-related macular degeneration — a phase IV randomised clinical trial with ranibizumab: the FLUID study. *BMC Oph-thalmol.* 2016;16:31. DOI: 10.1186/s12886-016-0207-3

3. Schmidt-Erfurth U, Chong V, Loewenstein A, et al. Guidelines for the management of neovascular age-related macular degeneration by the European Society of Retina Specialists (EURETINA). *Br J Ophthalmol.* 2014;98(9):1144–1167. DOI: 10.1136/bjophthal-mol-2014–305702

4. Wykoff CC, Clark WL, Nielsen JS, et al. Optimizing Anti-VEGF Treatment Outcomes for Patients with Neovascular Age-Related Macular Degeneration. *J Manag Care Spec Pharm.* 2018;24 (2-a Suppl): S3–S15. DOI: 10.18553/jmcp.2018.24.2-a.s3

5. Rosenfeld PJ, Puliafito CA, Michels S, et al. Systemic Bevacizumab (Avastin®) Therapy for Neovascular Age-Related Macular Degeneration (SANA) Study: Visual Acuity Outcomes. *Invest. Ophthalmol. Vis. Sci.* 2005;46(13):2310.

6. Emerson MV, Lauer AK, Flaxel CJ, et al. Intravitreal bevacizumab (Avastin) treatment of neovascular age-related macular degeneration. *Retina*. 2007;27(4):439–444. DOI: 10.1097/IAE.0b013e31804b3e15

7. Rich RM, Rosenfeld PJ, Puliafito CA, et al. Short-term safety and efficacy of intravitreal bevacizumab (Avastin) for neovascular age-related macular degeneration. *Retina*. 2006;26(5):495–511. DOI: 10.1097/01.iae.0000225766.75009.3a

8. Rosenfeld PJ, Brown DM, Heier JS, et al. Ranibizumab for neovascular age-related macular degeneration. *N Engl J Med.* 2006;355(14):1419–1431. DOI:10.1056/NEJMoa054481

9. Brown DM, Kaiser PK, Michels M, et al. Ranibizumab versus verteporfin for neovascular age-related macular degeneration. *N Engl J Med.* 2006;355(14):1432–1444. DOI: 10.1056/ NEJMoa062655

10. Heier JS, Boyer DS, Ciulla TA, et al. Ranibizumab combined with verteporfin photodynamic therapy in neovascular age-related macular degeneration: year 1 results of the FOCUS Study. *Arch Ophthalmol.* 2006;124(11):1532–1542. DOI: 10.1001/archopht.124.11.1532

11. Takeda AL, Colquitt J, Clegg AJ, Jones J. Pegaptanib and ranibizumab for neovascular age-related macular degeneration: a systematic review. *Br J Ophthalmol.* 2007;91(9):1177–1182. DOI: 10.1136/ bjo.2007.118562

12. Heier JS, Brown DM, Chong V, et al. Intravitreal aflibercept (VEGF trap-eye) in wet age-related macular degeneration. *Ophthalmology*. 2012;119(12):2537–2548. DOI: 10.1016/j.ophtha.2012.09.006

true. However this does not bother to perform post-registration trials later, which would allow receiving additional information upon brolucizumab therapy in the group of patients with a non-optimal nAMD control by anatomical points (clinicaltrials.gov identificator NCT04264819).

13. Mauri L, D'Agostino RB Sr. Challenges in the Design and Interpretation of Noninferiority Trials. *N Engl J Med.* 2017;377(14): 1357–1367. DOI: 10.1056/NEJMra1510063

14. Wickremasinghe SS, Janakan V, Sandhu SS, et al. Implication of recurrent or retained fluid on optical coherence tomography for visual acuity during active treatment of neovascular age-related macular degeneration with a treat and extend protocol. *Retina*. 2016;36(7):1331–1339. DOI: 10.1097/IAE.0000000000000902

15. Nguyen QD, Das A, Do DV, et al. Brolucizumab: Evolution through Preclinical and Clinical Studies and the Implications for the Management of Neovascular Age-Related Macular Degeneration. *Ophthalmology*. 2020;127(7):963–976. DOI: 10.1016/j.ophtha.2019.12.031

16. Dugel PU, Jaffe GJ, Sallstig P, et al. Brolucizumab Versus Aflibercept in Participants with Neovascular Age-Related Macular Degeneration: A Randomized Trial. *Ophthalmology*. 2017;124(9): 1296–1304. DOI: 10.1016/j.ophtha.2017.03.057

17. Cao J, Zhang S. Multiple comparison procedures. *JAMA*. 2014;312(5):543–544. DOI: 10.1001/jama.2014.9440

18. Dugel PU, Koh A, Ogura Y, et al. HAWK and HARRIER: Phase 3, Multicenter, Randomized, Double-Masked Trials of Brolucizumab for Neovascular Age-Related Macular Degeneration. *Ophthalmology*. 2020;127(1):72–84. DOI: 10.1016/j.ophtha.2019.04.017

19. Agostini H, Mulyukov Z, Tsilimbaris M, et al. Comparison of the Efficacy of Brolucizumab with Natural Disease Progression in Wet AMD Using Clinical Data from the Phase III HAWK and HARRIER Trials and Modelled Placebo Data. *Curr Eye Res.* 2020;45(10): 1298–1301. DOI: 10.1080/02713683.2020.1731832

20. Ho AC, Busbee BG, Regillo CD, et al. Twenty-four-month efficacy and safety of 0.5 mg or 2.0 mg ranibizumab in patients with subfoveal neovascular age-related macular degeneration. *Ophthalmology*. 2014;121(11):2181–2192. DOI: 10.1016/j.ophtha.2014.05.009

21. Yadav K, Lewis RJ. Gatekeeping Strategies for Avoiding False-Positive Results in Clinical Trials With Many Comparisons. *JAMA*. 2017;318(14):1385–1386. DOI: 10.1001/jama.2017.13276

22. Kertes PJ, Galic IJ, Greve M, et al. Canadian Treat-and-Extend Analysis Trial with Ranibizumab in Patients with Neovascular Age-Related Macular Disease: One-Year Results of the Randomized Canadian Treat-and-Extend Analysis Trial with Ranibizumab Study. *Ophthalmology*. 2019;126(6):841–848. DOI: 10.1016/j.ophtha.2019.01.013
23. Ross AH, Downey L, Devonport H, et al. Recommendations by a UK expert panel on an aflibercept treat-and-extend pathway for the treatment of neovascular age-related macular degeneration. *Eye (Lond)*. 2020;34(10):1825–1834. DOI: 10.1038/s41433-019-0747-x

24. Dugel PU, Singh RP, Koh A, et al. HAWK and HARRIER: Ninety-Six-Week Outcomes from the Phase 3 Trials of Brolucizumab for Neovascular Age-Related Macular Degeneration. *Ophthalmology*. 2021;128(1):89–99. DOI: 10.1016/j.ophtha.2020.06.028

СПИСОК ЛИТЕРАТУРЫ

1. DiMasi J.A., Grabowski H.G., Hansen R.W. Innovation in the pharmaceutical industry: New estimates of R&D costs // J Health Econ. 2016. Vol. 47. P. 20–33. DOI: 10.1016/j.jhealeco.2016.01.012

2. Arnold J.J., Markey C.M., Kurstjens N.P., Guymer R.H. The role of sub-retinal fluid in determining treatment outcomes in patients with neovascular age-related macular degeneration – a phase IV randomised clinical trial with ranibizumab: the FLUID study // BMC Ophthalmol. 2016. Vol. 16. P. 31. DOI: 10.1186/s12886-016-0207-3

3. Schmidt-Erfurth U., Chong V., Loewenstein A., et al. Guidelines for the management of neovascular age-related macular degeneration by the European Society of Retina Specialists (EURETINA) // Br J Ophthalmol. 2014. Vol. 98, No. 9. P. 1144–1167. DOI:10.1136/bjoph-thalmol-2014–305702

4. Wykoff C.C., Clark W.L., Nielsen J.S., et al. Optimizing Anti-VEGF Treatment Outcomes for Patients with Neovascular Age-Related Macular Degeneration // J Manag Care Spec Pharm. 2018. Vol. 24, (2-a Suppl) P. S3–S15. DOI: 10.18553/jmcp.2018.24.2-a.s3

5. Rosenfeld P.J., Puliafito C.A., Michels S., et al. Systemic Bevacizumab (Avastin®) Therapy for Neovascular Age–Related Macular Degeneration (SANA) Study: Visual Acuity Outcomes // *Invest. Ophthalmol. Vis. Sci.* 2005. Vol. 46, No. 13. P. 2310.

6. Emerson M.V., Lauer A.K., Flaxel C.J., et al. Intravitreal bevacizumab (Avastin) treatment of neovascular age-related macular degeneration // Retina. 2007. Vol. 27, No. 4. P. 439–444. DOI: 10.1097/ IAE.0b013e31804b3e15

7. Rich R.M., Rosenfeld P.J., Puliafito C.A., et al. Short-term safety and efficacy of intravitreal bevacizumab (Avastin) for neovascular age-related macular degeneration // Retina. 2006. Vol. 26, No. 5. P. 495–511. DOI: 10.1097/01.iae.0000225766.75009.3a

8. Rosenfeld P.J., Brown D.M., Heier J.S., et al. Ranibizumab for neovascular age-related macular degeneration // N Engl J Med. 2006. Vol. 355, No. 14. P. 1419–1431. DOI:10.1056/NEJMoa054481

9. Brown D.M., Kaiser P.K., Michels M., et al. Ranibizumab versus verteporfin for neovascular age-related macular degeneration // N Engl J Med. 2006. Vol. 355, No. 14. P. 1432–1444. DOI: 10.1056/ NEJMoa062655

10. Heier J.S., Boyer D.S., Ciulla T.A., et al. Ranibizumab combined with verteporfin photodynamic therapy in neovascular age-related macular degeneration: year 1 results of the FOCUS Study // Arch Ophthalmol. 2006. Vol. 124, No. 11. P. 1532–1542. DOI: 10.1001/ar-chopht.124.11.1532

11. Takeda A.L., Colquitt J., Clegg A.J., Jones J. Pegaptanib and ranibizumab for neovascular age-related macular degeneration: a systematic review // Br J Ophthalmol. 2007. Vol. 91, No. 9. P. 1177–1182. DOI: 10.1136/bjo.2007.118562

12. Heier J.S., Brown D.M., Chong V., et al. Intravitreal aflibercept (VEGF trap-eye) in wet age-related macular degeneration // Oph-thalmology. 2012. Vol. 119, No. 12. P. 2537–2548. DOI: 10.1016/j. ophtha.2012.09.006

13. Mauri L., D'Agostino Sr R.B. Challenges in the Design and Interpretation of Noninferiority Trials // N Engl J Med. 2017. Vol. 377, No. 14. P. 1357–1367. DOI: 10.1056/NEJMra1510063

14. Wickremasinghe S.S., Janakan V., Sandhu S.S. Implication of recurrent or retained fluid on optical coherence to-mography for visual acuity during active treatment of neovascular age-related macular degeneration with a treat and extend protocol // Retina. 2016. Vol. 36, No. 7. P. 1331–1339. DOI: 10.1097/IAE.00000000000002

15. Nguyen Q.D., Das A., Do D.V., et al. Brolucizumab: Evolution through Preclinical and Clinical Studies and the Implications for the Management of Neovascular Age-Related Macular Degeneration // Ophthalmology. 2020. Vol. 127, No. 7. P. 963–976. DOI: 10.1016/j.ophtha.2019.12.031

16. Dugel P.U., Jaffe G.J., Sallstig P., et al. Brolucizumab Versus Aflibercept in Participants with Neovascular Age-Related Macular Degeneration: A Randomized Trial // Ophthalmology. 2017. Vol. 124, No. 9. P. 1296–1304. DOI: 10.1016/j.ophtha.2017.03.057

17. Cao J., Zhang S. Multiple comparison procedures // JAMA. 2014. Vol. 312, No. 5. P. 543–544. DOI: 10.1001/jama.2014.9440

18. Dugel P.U., Koh A., Ogura Y., et al. HAWK and HARRIER: Phase 3, Multicenter, Randomized, Double-Masked Trials of Brolucizumab for Neovascular Age-Related Macular Degeneration // Ophthalmology. 2020. Vol. 127, No. 1. P. 72–84. DOI: 10.1016/j.ophtha.2019.04.017

19. Agostini H., Mulyukov Z., Tsilimbaris M., et al. Comparison of the Efficacy of Brolucizumab with Natural Disease Progression in Wet AMD Using Clinical Data from the Phase III HAWK and HARRIER Trials and Modelled Placebo Data // Curr Eye Res. 2020. Vol. 45, No. 10. P. 1298–1301. DOI: 10.1080/02713683.2020.1731832

20. Ho A.C., Busbee B.G., Regillo C.D., et al. Twenty-four-month efficacy and safety of 0.5 mg or 2.0 mg ranibizumab in patients with subfoveal neovascular age-related macular degeneration // Oph-thalmology. 2014. Vol. 121, No. 11. P. 2181–2192. DOI: 10.1016/j. ophtha.2014.05.009

21. Yadav K., Lewis R.J. Gatekeeping Strategies for Avoiding False-Positive Results in Clinical Trials With Many Comparisons // JAMA. 2017. Vol. 318, No. 14. P. 1385–1386. DOI: 10.1001/jama.2017.13276
22. Kertes P.J., Galic I.J., Greve M., et al. Canadian Treat-and-Extend Analysis Trial with Ranibizumab in Patients with Neovascular Age-Related Macular Disease: One-Year Results of the Randomized Canadian Treat-and-Extend Analysis Trial with Ranibizumab Study // Ophthalmology. 2019. Vol. 126, No. 6. P. 841–848. DOI: 10.1016/j. ophtha.2019.01.013

23. Ross A.H., Downey L., Devonport H., et al. Recommendations by a UK expert panel on an aflibercept treat-and-extend pathway for the treatment of neovascular age-related macular degeneration // Eye (Lond). 2020. Vol. 34, No. 10. P. 1825–1834. DOI: 10.1038/s41433-019-0747-x

24. Dugel P.U., Singh R.P., Koh A., et al. HAWK and HARRIER: Ninety-Six-Week Outcomes from the Phase 3 Trials of Brolucizumab for Neovascular Age-Related Macular Degeneration // Ophthalmology. 2021. Vol. 128, No. 1. P. 89–99. DOI: 10.1016/j.ophtha.2020.06.028

DOI: https://doi.org/10.17816/0V63571

AUTHORS' INFO

*Sergey Yu. Astakhov, Dr. Sci. (Med.), professor; address: 6–8 L'va Tolstogo str., Saint Petersburg, 197022, Russia; Phone: +7(812)338-71-73; ORCID: http://orcid.org/0000-0003-0777-4861; eLibrary SPIN: 7732-1150; e-mail: astakhov0761@mail.ru

Svyatoslav L. Plavinskii, MD, PhD, Dr. Sci (Med), professor; ORCID: http://orcid.org/0000-0001-9159-6177; eLibrary SPIN: 5660-4661; e-mail: s.plavinskij@szgmu.ru

* Corresponding author / Автор, ответственный за переписку

ОБ АВТОРАХ

*Сергей Юрьевич Астахов, д-р мед. наук, профессор; адрес: Россия, 197022, Санкт-Петербург, ул. Льва Толстого, д. 6–8; телефон: +7(812)338-71-73; ORCID: http://orcid.org/0000-0003-0777-4861; eLibrary SPIN: 7732-1150; e-mail: astakhov0761@mail.ru

Святослав Леонидович Плавинский, д-р мед. наук, профессор; ORCID: http://orcid.org/0000-0001-9159-6177; eLibrary SPIN: 5660-4661; e-mail: s.plavinskij@szgmu.ru